Unix Systems: Configuring X

Bob Dowling^�

August 23, 1996

Course notes

Warning: In order to explain something as immense as X and to still say something useful the author has been forced to take some liberties with details. Put more bluntly the author is going to lie through his teeth in this document. It is hoped and intended that none of the deceits will prove to be dangerous ones.

�University of Cambridge Computing Service, rjd4@ucs.cam.ac.uk 1

Contents

1 Introduction 7 1.1 What this course covers : : : : : : : : : : : : : : : : : : : : : 7 1.2 Prerequisites for This Course : : : : : : : : : : : : : : : : : : 7

2 How to access X 8 2.1 On a personal workstation starting with X running : : : : : : : 8 2.2 On a personal workstation starting without X running : : : : : 8 2.3 On a dedicated X-terminal : : : : : : : : : : : : : : : : : : : : 9 2.4 On an IBM PC on the PWF : : : : : : : : : : : : : : : : : : : 9

3 The components of an X system 10 3.1 The X server : : : : : : : : : : : : : : : : : : : : : : : : : : : : 10 3.2 The X clients : : : : : : : : : : : : : : : : : : : : : : : : : : : 11 3.3 The X11 protocol : : : : : : : : : : : : : : : : : : : : : : : : : 11

4 X displays 12 4.1 The display parameter : : : : : : : : : : : : : : : : : : : : : : 12 4.2 Telling the client : : : : : : : : : : : : : : : : : : : : : : : : : 13 4.3 X permission mechanisms : : : : : : : : : : : : : : : : : : : : : 13 4.3.1 Host-based permissions : : : : : : : : : : : : : : : : : : 13

4.3.2 Token-based permissions : : : : : : : : : : : : : : : : : 15 4.4 Troubleshooting : : : : : : : : : : : : : : : : : : : : : : : : : : 16 4.4.1 No display : : : : : : : : : : : : : : : : : : : : : : : : : 16 4.4.2 Permission denied : : : : : : : : : : : : : : : : : : : : : 16 4.5 Starting X clients on remote machines : : : : : : : : : : : : : : 17

5 Some commonly used X clients 19 5.1 xbiff : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 19 5.2 xclock : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 19 5.3 xterm : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 20 5.3.1 Main options menu : : : : : : : : : : : : : : : : : : : : 21 5.3.2 VT options menu : : : : : : : : : : : : : : : : : : : : : 22 5.3.3 VT fonts menu : : : : : : : : : : : : : : : : : : : : : : 23 5.3.4 Cut and Paste : : : : : : : : : : : : : : : : : : : : : : : 24 5.3.5 Running other programs in xterm : : : : : : : : : : : : 24

6 Window Managers 26 6.1 Why have window managers? : : : : : : : : : : : : : : : : : : 26 6.2 What window managers provide : : : : : : : : : : : : : : : : : 26 6.3 How to customize twm : : : : : : : : : : : : : : : : : : : : : : 30

7 X startup 37

7.1 X startup with an X-terminal : : : : : : : : : : : : : : : : : : : 37 7.2 X startup from text mode : : : : : : : : : : : : : : : : : : : : : 38 7.3 The .xsession/.xinitrc file : : : : : : : : : : : : : : : : : : 39 7.4 Troubleshooting : : : : : : : : : : : : : : : : : : : : : : : : : : 40

8 Structure of a (fairly typical) X client 42

9 Configuring an X client 45 9.1 Resources : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 45 9.2 Setting resource values : : : : : : : : : : : : : : : : : : : : : : 45 9.3 Using xrdb : : : : : : : : : : : : : : : : : : : : : : : : : : : : 46 9.3.1 Using xrdb to query the server for resources : : : : : : 46 9.3.2 Using xrdb to add resources to the server : : : : : : : : 47 9.3.3 Using xrdb to set the server’s resources : : : : : : : : : 48 9.3.4 xrdb at start-up : : : : : : : : : : : : : : : : : : : : : : 48 9.4 X client geometry : : : : : : : : : : : : : : : : : : : : : : : : : 50 9.4.1 Position : : : : : : : : : : : : : : : : : : : : : : : : : : 50 9.4.2 Size : : : : : : : : : : : : : : : : : : : : : : : : : : : : 51 9.4.3 Geometry resources : : : : : : : : : : : : : : : : : : : : 51 9.5 Colours : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 52 9.6 Fonts : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : : 52

9.6.1 X Logical Font Descriptor Structure : : : : : : : : : : : 54 9.6.2 Wildcarding in XLFDs : : : : : : : : : : : : : : : : : : 56 9.7 Using instance names to select resources : : : : : : : : : : : : 58

LIST OF FIGURES 6

List of Figures

1 xbiff’s two states : : : : : : : : : : : : : : : : : : : : : : : : 19 2 xclock — analogue and digital : : : : : : : : : : : : : : : : : 20 3 An xterm with scrollbar and default geometry : : : : : : : : : 21 4 An xterm decorated by twm : : : : : : : : : : : : : : : : : : : 27 5 A collection of decorations : : : : : : : : : : : : : : : : : : : : 28 6 xedit applied to /etc/motd : : : : : : : : : : : : : : : : : : : 42 7 Widget tree of xedit labelled by class name : : : : : : : : : : 43 8 Widget tree of xedit labelled by instance name : : : : : : : : 44 9 Effects of sign in the geometry specification : : : : : : : : : : 51 10 Some lines from rgb.txt : : : : : : : : : : : : : : : : : : : : : 53 11 The xfontsel client : : : : : : : : : : : : : : : : : : : : : : : 57

1 INTRODUCTION 7

1 Introduction

1.1 What this course covers

ffl Defining X ffl Starting up X ffl Common problems with X ffl Common X tools ffl Configuring X

1.2 Prerequisites for This Course

ffl Some experience with X ffl Some familiarity with shell syntax ffl Basic familiarity with Unix